



Article Environmental Performance of the Tourism Sector from a Gender Diversity Perspective

Yakira Fernández-Torres ¹, Milagros Gutiérrez-Fernández ¹ and Clara Gallego-Sosa ^{2,*}

- ¹ Department of Financial Economics and Accounting, Faculty of Business, Finance and Tourism,
- University of Extremadura, 10071 Cáceres, Spain; yakiraft@unex.es (Y.F.-T.); mgutierrezf@unex.es (M.G.-F.)
 ² Department of Economics, Faculty of Economics and Business, University of Extremadura,
 - 06006 Badajoz, Spain
- * Correspondence: clarags@unex.es

Abstract: The tourism sector is a driver of economic development characterised by its environmental impact. It is a prevalent part of the 2030 Agenda, given its potential to help meet the Sustainable Development Goals (SDGs). At the same time, board gender diversity is considered essential for companies to implement environmentally sustainable initiatives. However, analysis of the relationship between the role of women on boards and environmental performance has been neglected in the tourism literature. This paper adopts a novel approach to the study of this sector by analysing the relationship between gender diversity on the board of directors and companies' environmental practices. A fixed effects model is estimated using an international sample of 120 listed tourism companies for the period 2002 to 2019. The results show that boards that are more gender diverse and have a greater female presence are associated with poorer environmental performance and a weaker implementation of policies and practices to reduce resource use and emissions. However, board gender diversity aids performance in environmental innovation.

Keywords: Sustainable Development Goals (SDGs); environmental performance; gender diversity; board of directors; tourism sector; critical mass

1. Introduction

In September 2015, the United Nations Assembly approved the 2030 Agenda, defined as a 'plan of action for people, planet and prosperity' [1] (p. 1). Its aim is to contribute to the sustainable development of the planet by targeting 17 major challenges, formalised as the Sustainable Development Goals (SDGs). Most fall within one of the three dimensions of corporate social responsibility (CSR): the economic, the social and the environmental. Achieving these goals requires effective joint action by governments, civil society and the private sector [1]. Focusing on the role of business, tourism plays a relevant role in the 2030 Agenda. Notably, it is dealt with in 3 of the 17 SDGs: SDG 8 (Decent Work and Economic Growth), SDG 12 (Responsible Consumption and Production) and SDG 14 (Life Below Water). These SDGs are directly linked to the tourism sector because of this sector's capacity to contribute actively to meeting these goals. From an economic perspective (such as SDG 8), tourism is a key sector for socio-economic progress given its contribution to employment [2,3], the economy, international trade and infrastructures [4,5]. However, the activity of tourism companies also raises considerable challenges that must be addressed (see SDGs 12 and 14). Several studies have reported the adverse effects of tourism on the environment in the form of, for example, greenhouse gas (GHG) emissions, which cause climate change [6-8]. It is also estimated that the rapid growth of the sector may double the use of natural resources from 2010 to 2050 [9].

Given the interrelationships among the SDGs, tourism activity indirectly affects other challenges in addition to those mentioned earlier. For example, according to Borja et al. [10], the environmental impact of tourism will have repercussions for public well-being and



Citation: Fernández-Torres, Y.; Gutiérrez-Fernández, M.; Gallego-Sosa, C. Environmental Performance of the Tourism Sector from a Gender Diversity Perspective. *Int. J. Environ. Res. Public Health* **2021**, *18*, 8834. https://doi.org/10.3390/ ijerph18168834

Academic Editors: Heesup Han and Paul B. Tchounwou

Received: 30 June 2021 Accepted: 18 August 2021 Published: 22 August 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). health, an issue covered by SDG 3 (Good Health and Well-being). The negative role of tourism in this sense stems from the fact that environmental pollution causes cancer [11], respiratory problems and cardiovascular disease [12] and affects the nervous system, increasing stress [13], among other effects. Consequently, the relationship between tourism and climate change has become an intensely debated issue in recent times [8].

From a social perspective, the potential of the tourism sector to create employment [3] means that it can contribute to meeting SDG 5 (Gender Equality), which aims to achieve gender equality and empower women and girls [1] through decent work opportunities for women [14]. At the management level, although the glass ceiling that prevents women from climbing the corporate ladder has weakened in recent decades [15] and a higher proportion of women now occupy management positions [16], women still face difficulties in their career development in the tourism sector. This situation highlights the need to address this issue.

Linked to the previous discussion, there is a close relationship between gender diversity in the tourism sector and the environment. Adopting a gender focus to understand the differentiating impact of tourism in terms of sustainability is crucial [17]. Changes in the environmental sustainability of companies can be explained by the potential influence of women board members in this area [18]. Therefore, achieving gender diversity on the board of directors is essential for companies to undertake environmental actions [19].

However, despite the extensive literature on tourism in recent decades, research on gender diversity in this sector and its consequences remains a minor topic [20,21]. Moreover, studies of tourism and climate change from a gender perspective are scarce [22], highlighting the importance of research in this area [23]. Specifically, only one study seems to have examined the relationship between gender diversity and CSR in terms of each of its three dimensions (economic, social and environmental) in tourism companies [24]. The study showed a positive influence of board gender diversity on environmental performance. However, the literature on the role of gender diversity in environmental performance includes studies of other sectors, although these studies reveal a lack of consensus in their findings. While numerous studies suggest the positive impact of actions by women directors in this area [25–27], others indicate that board gender diversity is not a differentiating factor in terms of companies' environmental performance [28,29]. Some studies even suggest that organisations with higher female representation on their boards have inferior environmental performance [30,31].

To summarise, the discussion so far highlights the importance of the tourism sector as a driver of economic development and a potential agent of change in the fight against climate change, the key role of gender diversity in environmental sustainability and the virtually non-existent evidence of this relationship in tourism companies. These are the principal motivations for this research. The aim of this study is to analyse the influence of tourism companies' board gender diversity on these firms' environmental practices. A fixed effects model was estimated using an international sample of 120 listed tourism companies for the period 2002 to 2019.

This research makes several contributions to the literature. First, this study offers what seems to be the first ever analysis of the influence of board gender diversity on environmental performance that focuses exclusively on the tourism sector. A further novel feature of the study is the use of not only a general environmental performance indicator as the dependent variable but also indicators for each of the three dimensions of environmental performance: reduction of resource use, reduction of emissions and innovation. Given the distinctive characteristics of this sector and its strategic role in the fight against climate change, evidence of the factors that condition corporate governance and the consequences for firms' environmental performance is valuable. This study thus confirms the importance of considering gender-related issues when explaining initiatives by tourism companies to help care for the planet. The study also shows the importance of reaching a critical mass of women on the board of directors in order for their presence to influence environmental policy. This theory has been virtually ignored in studies of gender

diversity and environmental sustainability [32]. Finally, the inclusion of a larger number of gender variables than is usual in the literature makes the study novel and relevant.

This paper is divided into six sections, the first being this introduction. Next, Section 2 offers a literature review, summarising the existing theories that support the relationship of interest, as well as evidence of the role of gender diversity in corporate environmental performance. Section 3 describes the sample selection and choice of variables. Section 4 outlines the method. Section 5 presents and discusses the results. Finally, Section 6 presents the conclusions, limitations and possible future lines of research.

2. Theoretical Framework

2.1. Theories of the Role of Women on the Board of Directors and the Influence of Women on CSR

The relationship between the presence of women on the board of directors and CSR performance is supported by several theories. According to these theories, gender diversity contributes to the effectiveness of boards of directors. This position is supported by various arguments used to provide a theoretical foundation in the literature [33–35]. Hence, these theories, which are described below, are used as the basis for this paper.

The first, resource dependence theory [36], is based on the notion that companies act within an open system, making them dependent on their environment. In line with this idea, the board of directors is an important part of corporate performance because it plays the role of intermediary between organisations and their context. From this viewpoint, the board performs two essential functions. First, by opening information channels between the company and its environment, it reduces the uncertainty under which the company operates. Second, the board provides critical organisational resources such as experience and knowledge [37,38]. Through these two functions, gender diversity plays an essential role in company performance for several reasons. Specifically, having women on the board provides new ways to access resources and information [39], greater cognitive diversity [40], different points of view and greater policy options [41]. These outcomes of gender diversity result in more effective decision making [33,42], which leads to better board performance in a variety of areas, including CSR [43]. In addition, women have greater awareness and experience in social and environmental issues [34], further supporting the arguments that underpin the results of studies based on resource dependence theory. These results indicate that companies with higher female representation on their boards have better CSR practices and environmental performance [26,44].

The second theory, stakeholder theory [45], presents the idea that a company cannot focus solely on benefiting shareholders. Instead, it must address the interests of other stakeholders that are also affected by its actions and can in turn affect its activity. Examples include employees, society and the environment. From this perspective, women have been found to have a stronger social orientation than men [46]. Together with a more inclusive personality, this orientation leads to greater concern for stakeholders and attention to their needs [47], resulting in a greater number of social and environmental practices to meet stakeholder expectations [34,48]. Several studies support this theory in terms of the positive effect of gender diversity on environmental performance [49–51].

These organisational theories are complemented by social role theory [52], which is based on the idea that behaviour differs between men and women because it is determined by expectations according to the gender roles within a given society [52,53]. For example, the female gender is associated with the role of caring for children and housework. This role means that, in general, women are more understanding and attentive than men [54]. Women are also considered more collaborative [55], democratic, participative and interactive [33], which enables them to communicate more effectively with their subordinates [56] and give a voice to minority groups, aiding their empowerment [43]. In addition, women are more transparent because they advocate greater disclosure of practices, such as social responsibility practices [57,58]. These characteristics of the female gender explain why leadership styles differ between men and women. Accordingly, the effect of corporate governance on company performance is conditioned by gender diversity among directors [19].

In relation to these arguments, critical mass theory also refers to the number of women on the board of directors [59]. Under this theory, the inclusion of women on the board may be a merely symbolic gesture to comply with regulations and improve an organisation's image [60]. Thus, the female gender, which is considered a token, would not improve the effectiveness of the board of directors [61]. Consequently, it would not influence the company's CSR practices either [62]. Based on these arguments, the concept of tokenism means that women can only provide new points of view and improve a group's actions, such as by increasing social and environmental responsibility, if there is a minimum representation of women in that group, known as a critical mass [63–65]. Kanter [59] defined four types of groups depending on the percentage of minority members: uniform (absence of minority members), skewed (less than 20% minority members), tilted (20% to 40% minority members) and balanced (40% to 60% minority members). According to the classification proposed by Kanter [59] it is essential to achieve a "tilted" group by having at least 30% women on the board of directors [66], which is equivalent in absolute terms to three women [67], to bring about significant, perceptible changes in group dynamics [68]. Based on these arguments, several studies have confirmed the need for representation of at least three women on the board in order for their presence to contribute positively to socially responsible actions [69] such as environmental initiatives and programmes [70]. However, some scholars disagree, reporting that reaching a critical mass of two women is sufficient to reduce GHG emissions [71] and ensure greater environmental innovation [72].

This section addresses the main theories underpinning the relationship of interest in this research. The next section discusses the evidence of the role of gender diversity in corporate environmental performance.

2.2. Board Gender Diversity and Environmental Performance

Multiple studies have shown that board gender diversity plays an essential role in a company's environmental performance [50,73]. These studies are based on differences in behaviour between men and women. Specifically, women are more aware of the exploitation of the planet [74], more sensitive to environmental problems and more likely to tackle the needs of stakeholders [43,75,76]. Therefore, the presence of women on the board shifts the orientation of companies towards tackling environmental sustainability [18] and meeting SDG 13 (Climate Action) [77]. It also contributes to the implementation of environmental policies and initiatives [78–80] and green entrepreneurship programmes [81], all of which results in improved environmental performance [82].

However, the literature is far from offering a consensus on the positive influence of gender diversity on environmental performance. This lack of consensus is reflected in Table 1, which summarises the empirical evidence reported in previous studies. Table 1 shows which type of environmental performance variable was used (a general indicator or a specific indicator for a given environmental dimension). The results for the relationship of interest are also shown.

Authors	Performance Variable	Results
[18,24–27,32,42–44,47,50,51,64,70,73,79,82–88] [27,47,85,89] [27,71,85] [19,47,72,85,90–92]	Environmental performance Use of resources Emissions Environmental innovation	Positive relationship
[30,31,93,94] [87] [95]	Environmental performance Consumption of water and energy Environmental innovation	Negative relationship
[28,34,96–106] [29] [27,85]	Environmental performance Consumption of energy Environmental innovation	No relationship

Table 1. Review of studies of the relationship between gender diversity and environmental performance.

Source: authors, based on the literature review.

Assuming that the leader's gender determines performance in terms of corporate practices, Table 1 shows a number of studies in which board gender diversity is identified as central for enhancing environmental performance (positive relationship). This relationship has been confirmed for samples from various countries and regions. Specifically, studies have been conducted in the United States [26,42,43,50,64,83], France [47], the United Kingdom [25], China [86], Australia [85], Europe [27,77,88] and globally [44,73,96].

Furthermore, the positive influence of board gender diversity on environmental performance has also been corroborated for various sectors, even though sector is a conditioning factor for the sign and significance of the causal relationship between gender and environmental performance [30]. Specific sectors where such research has been conducted include tourism [24], banking [32], energy [51], oil and gas [82] and manufacturing [106]. In the most polluting sectors, the presence of female board members has a more pronounced effect on the adoption of environmental policies [26,79] and environmental innovation [72,91,92].

Besides showing that greater inclusion of women on the board positively influences environmental performance in general, some studies have focused on specific aspects of the environment. For example, Biswas et al. [85] found that companies with greater board gender diversity implement policies and take actions related to reducing resource use and emissions, as well as environmental product and service innovation. Similarly, Orazalin and Baydauletov [27] and Burkhardt et al. [47] have corroborated the positive influence of female representation on the reduction of resource use. The contribution of women directors to emissions reduction policies [27] and environmental innovation [47] has also been reported.

Several authors also claim that although the presence of women directors leads to the implementation of social practices, the same cannot be said of environmental performance. For instance, Cuadrado-Ballesteros et al. [104] and Beji et al. [105] have suggested that although board gender diversity positively influences CSR practices, it does not positively influence the environmental dimension of CSR. Alazzani et al. [103] reported that while board gender diversity gives rise to improved social action by firms, the same is not true in relation to the environment. Such findings have been corroborated by studies that exclusively examine environmental performance [97,106].

Focusing on specific issues, female representation on the board does not influence the company's approach to climate change [101]. Moreover, it does not lead to greater environmental awareness [100] or determine energy use [29]. This lack of female influence on corporate action may owe to the under-representation of women on the board [28,95,97] or to resistance that reduces women's voice in decision making because of the stereotype of men as the dominant gender [107].

Finally, some studies have shown a negative relationship between these two variables. However, there are considerably fewer than those showing a positive or neutral relationship, as reflected by Table 1. For example, Reyes-Bastidas and Briano-Turrent [93] reported that while the presence of women on the board favours performance in terms of the economic dimension of CSR, it negatively influences performance in terms of the social and environmental dimensions. Deschênes et al. [30] reported that companies with greater female representation on the board have better CSR practices, but not in the environmental sphere.

Ardito et al. [31] and Shu and Chiang [94] also found that as the number of women on the board of directors increases, environmental performance worsens. These results can be justified by two lines of argument. First, women may be welcomed onto the board of directors merely to improve the company's image, without any real interest in environmentally friendly practices [30]. Second, communication problems and interpersonal conflicts among board members may emerge due to greater diversity [108]. These issues can result in poorer decision making because it becomes more difficult to reach a consensus on critical decisions [109].

In conclusion, although some studies suggest the absence of a relationship between board gender diversity and corporate environmental performance, or even that this relationship is negative, the bulk of the literature provides evidence of the positive role of women directors in improving corporate environmental performance. Following this reasoning and the argument that women show greater concern for the environment and tend to take actions that are favourable, or at least less harmful, to the planet [110], the following research hypotheses are proposed:

Hypothesis 1 (H1). *The presence of women on the board contributes to better environmental performance in the tourism sector.*

Hypothesis 2 (H2). The presence of women on the board contributes to better environmental performance by reducing the use of resources in the tourism sector.

Hypothesis 3 (H3). The presence of women on the board contributes to better environmental performance by reducing emissions in the tourism sector.

Hypothesis 4 (H4). *The presence of women on the board contributes to better performance in terms of environmental innovation in the tourism sector.*

3. Data and Variables

This study used a sample of 120 tourism companies from 26 countries across five continents. The data spanned the period of 2002 to 2019. This period was chosen because these years corresponded to the earliest and most recent years for which environmental, social and governance (ESG) data were available at the time of data collection. These tourism companies are publicly listed, and on 25 March 2020, they had at least 4 years of available data. The selection criterion of publicly listed status was applied because data availability is better for listed companies. Moreover, the implementation of environmental initiatives requires substantial investment in technology, infrastructure and human resources, and listed companies have better access to funds for such practices than unlisted companies [111]. A minimum time span of 4 years was chosen because four periods are required to provide panel data that enable the use of lags of the explanatory variables as instruments. This feature is essential for the econometric model to verify the existence of endogeneity and to correct for this issue. In addition, this criterion also provided a more balanced panel.

The selected companies were drawn from an initial sample of 399 listed companies in the Thomson Reuters Eikon database [112]. Of these, 279 were discarded because they did not have at least 4 years of data on environmental performance. This data source was selected because it provides rich data covering financial as well as ESG data.

According to the sector classification proposed by Refinitiv [113], the sampled companies can be divided into four subsectors: restaurants and bars, leisure and recreation, casinos and gaming, and hotels, motels and cruises. Table 2 presents the distribution of the sampled companies by continent and country.

3.1. Dependent Variables

To measure the environmental performance of tourism companies, four indicators from the Thomson Reuters Eikon [112] database were used. This database provides scores on the three dimensions of ESG (aggregated indicators) and their underlying categories, as well as information on the items used to calculate these scores. Thanks to these rich data, studies such as those by Kyaw et al. [25] and Birindelli et al. [32] have used the score of the environmental and social dimensions as the dependent variable. The scores for the pillars of the aggregate measures have been used to a lesser extent in the CSR literature [47,51,85]. However, these pillars are important and should be considered, given that using only the aggregate indicators for each dimension could mean that relevant information is overlooked.

Therefore, in this study the dependent variables were taken as the overall environmental performance score and its three categories: resource use (20 items), emissions (28 items) and innovation (20 items). This approach lent robustness to the results, thanks to the use of various measures, while helping identify the importance of each of the areas of environmental performance. This study thus addresses the shortcomings mentioned earlier. To help identify the information covered by the environmental measures, Table 3 shows the labels assigned to these four variables, their definition and their content.

Continent	Country	Number of Companies
Africa		3
	South Africa	3
America		53
	Canada	4
	United States	49
Asia		25
	China	1
	Republic of Korea	2
	Philippines	1
	Hong Kong	7
	Japan	4
	Macao	3
	Malaysia	3
	Singapore	1
	Sri Lanka	1
	Thailand	1
	Taiwan	1
Europe		25
	Germany	1
	Spain	1
	France	3
	Gibraltar	1
	Greece	1
	Ireland	1
	Isle of Man	1
	Italy	1
	Malta	1
	United Kingdom	14
Oceania		14
	Australia	12
	New Zealand	2
Total		120

Table 2. Distribution of the sampled companies by continent and country.

Source: authors, based on Refinitiv [112].

Finally, regarding the method, calculation of the environmental score involved obtaining the relative value for each category and then calculating the overall index. Therefore, the starting point was to determine the score for each category as a weighted average of the groups of data for the industry that comprise it. Accordingly, the weighting for each group varied by industry. These data were previously generated by assigning a score to each of the areas measured by the category. Proxies were used where it was not possible to find data for a specific variable. The category scores were normalised and expressed as percentages ranging from 0 to 100. Finally, the score for the environmental pillar was calculated using the sum of the values resulting from the weightings of its three categories. The resource use and emissions categories were given a weighting of 35%, and the innovation category was given a weighting of 29% [114].

Label	Definition	Content
EnvSc	General environmental performance score	Uses information provided by each company on its ability to operate with a efficient use of natural resources, the reduction of emissions, and innovation ar support for the development of eco-efficient products and services
ResUseSc	Resource use score	Captures the performance and ability o each company to reduce the use of materials, energy and water and to finc more eco-efficient solutions by improvir supply chain management
EmiSc	Emissions reduction score	Measures the commitment and effectiveness of each company in reducir emissions in production and operationa processes
InnoSc	Innovation score	Captures the ability of each firm to reduce its environmental impact and create new market opportunities through new environmental technologies or processes ecologically designed products

 Table 3. Environmental performance variables.

Source: authors, based on Refinitiv [112].

Thus, the four environmental variables used in this study took values ranging from 0 to 100, where 0 indicates the worst possible environmental performance and 100 denotes the best possible environmental performance [87].

3.2. Independent Variables

This section presents and justifies the choice of independent variables. These variables can be divided into two groups. The first consists of variables of interest used to address the research aims. The second group consists of control variables included to ensure that the econometric model was correctly specified. The data used to create these variables were gathered from the Thomson Reuters Eikon database [112]. On the whole, the variables of interest for this study were the board gender diversity variables that have most commonly been used in studies of corporate environmental performance, as detailed in Table 4.

Label	Definition	Supporting Literature
Dum1	Dummy taking a value of 1 if there is at least one woman on the board of directors, and 0 otherwise	[18,25,72,80,85,89]
Dum2	Dummy taking a value of 1 if there are at least two women on the board of directors, and 0 otherwise	[58,72,89]
Dum3	Dummy taking a value of 1 if there are at least three women on the board of directors, and 0 otherwise	[18,32,70,89,115,116]
Dum30	Dummy taking a value of 1 if there are at least 30% of women on the board of directors, and 0 otherwise	[95]
Dum40	Dummy taking a value of 1 if there are at least 40% of women on the board of directors, and 0 otherwise	[95,117]
Nwom	Number of female board members	[29,31,64,72,84,89]
Pwom	Proportion of women directors, calculated as the ratio of the number of women on the board to the total number of board members	[24,27,47,50,51,73,85,87,97]
Blau	Index of gender diversity on the board of directors [118]	[26,47,80,103]

Table 4. Gender diversity variables.

Source: authors, based on a review of the studies cited in the table.

A large number of gender variables were selected. In fact, this study used more such variables than many other studies of the influence of board gender diversity on environmental performance [50,82,83]. This large number of variables lends robustness to the empirical evidence.

The rationale for using these gender measures is that several studies have reported that women are more likely to engage in socially responsible practices given their greater focus on stakeholders [73], and in particular, on environmental sustainability [18]. However, scholars disagree over the minimum number of women board members required in order for them to exert a noticeable influence on corporate performance [67,68], as described by critical mass theory [59]. These reasons, together with the limited evidence of the impact of a critical mass of women directors on environmental sustainability [32], motivated the decision to include the dichotomous variables *Dum1*, *Dum2* and *Dum3*.

The variable *Dum1* identifies whether merely having at least one woman director plays a relevant role in environmental performance. For example, Shoham et al. [18] showed that companies whose boards of directors had at least one woman were more predisposed towards environmental sustainability. Similarly, He and Jiang [72] and Atif et al. [89] have reported that having at least two women on the board of directors is sufficient for there to be a noticeable female influence on environmental performance and renewable energy consumption, respectively, hence the reason for including *Dum2*. However, a higher proportion of studies have shown that companies with at least three female directors can ensure that women have a stronger role on the board [57,115], pay more attention to environmental issues [64,70] and face less environmental litigation [115]. Hence, *Dum3* was included in the model. In addition, some authors, such as Kyaw et al. [25] and Shoham et al. [18], do not subscribe to critical mass theory. Instead, they suggest that a minimum level of female representation is not necessary for women to promote socially and environmentally responsible actions.

The variables *Dum30* and *Dum40* were included to control for the possibility of significant differences in business performance between companies with at least 30% and 40% of women directors, respectively, and companies that did not meet these thresholds. The aim was to explore the possible repercussions of corporate governance gender quotas, such as those proposed by the European Parliament and Council Directive [119]. Regarding the arguments in the literature, Ahern and Dittmar [120] reported that restructuring the board of directors by including women on the board to comply with gender quotas lowers the value of the company. However, according to Birindelli et al. [32], achieving 30% female representation on the board reflects a change in the trend of environmental performance, with the sign depending on the gender of the chief executive officer (CEO). This change is negative when the CEO is male, but positive when the CEO is female. Gallego-Sosa et al. [95] reported that achieving a minimum of 40% women directors does not affect environmental performance. In contrast, Lafuente and Vaillant [121] reported that a heterogeneous board in terms of gender, composed of 40% to 60% women (a balanced group according to Kanter), leads to better company performance. Crucially, these measures are novel in gender and corporate governance studies [117]. They have only been used in environmental research by Gallego-Sosa et al. [95].

The variables *Nwom* and *Pwom* are also essential. The variable *Nwom* was included because socially responsible companies have more women than non-socially responsible companies [31,62]. In addition, the number of women on the board of directors has been found to have a positive influence on environmental performance by increasing the implementation of environmental practices [64,84] such as those related to the use of renewable energy [89]. The level of female representation on the board of directors (*Pwom*) is also important because a higher proportion of women directors positively affects corporate environmental performance [24,25,87]. This influence is explained by the implementation of environmental strategies [86] and environmental practices [47]. Such environmental strategies include those related to the reduction of resource use and emissions [27,85], the

formation of strategic alliances in renewable energy [82] and environmental innovation measured in terms of patents [91].

Finally, the variable *Blau* was included. This index is the most widely used diversity index in the literature, with several studies citing it as a good measure of diversity [26,122]. Lu and Herremans [26] and Burkhardt et al. [47] reported that board gender diversity positively influences environmental performance. The values of the Blau index range from 0 to 0.5, where 0 indicates the lowest possible diversity due to the absence of women directors, and 0.5 indicates the maximum diversity when the number of female board members equals the number of male board members. This index was calculated using the following formula [118]:

Blau index =
$$1 - \sum_{i=1}^{n} P_i^2$$

where P_i is the percentage of men or women on the board of directors and n is the number of categories (i.e., male and female).

The second group of independent variables consists of the control variables, which are presented in Table 5. These variables were selected because their influence on environmental performance receives support in the literature, as indicated by the cited studies. Six control variables were used. Four were measures of corporate governance (*IndepMem, AveTen, EnvTra* and *Ndir*), and two were financial indicators, namely the size of the company (*Temp*) and financial performance (return on assets or *ROA*).

Label	Definition	Supporting Literature
IndepMem	Percentage of independent board members	[25,47,72,105]
AveTen	Average number of years that board members have held the position	[30,123,124]
EnvTra	Dummy taking a value of 1 if environmental management training policies are implemented, and 0 otherwise	[95]
Ndir	Number of board members	[26,102,103,116]
Temp	Number of full-time employees at the end of the tax period	[31,44,64,72,116]
ROA	Return on assets: ratio of profit after tax to average assets for the tax year	[26,64,102,111]

Table 5. Control variables.

Source: authors, based on a review of the studies cited in the table.

First, the degree of independence of the board was included in the model, measured as the percentage of independent directors (*IndepMem*). Several studies have shown the positive influence of board independence on the number of green initiatives [72] and environmental performance [47,111]. One explanation for this finding is that in addition to reducing agency costs, independent members have a greater focus on long-term actions, stakeholder needs [125] and environmental compliance [126].

The *AveTen* and *EnvTra* variables were included because of the scarcity of research on the influence of board members' experience on business performance [123] and the importance of environmental management training in corporate environmental action [95], respectively. Research is lacking in this area, even though board members' experience and education are closely related to business performance [127] and hence environmental performance [101].

The variable *Ndir* (i.e., the number of board members) is also important. Although some studies suggest that smaller boards are more effective, participatory and cohesive [128], larger boards are widely accepted to have greater diversity and access to information [129]. For instance, several studies have shown that board size positively

influences the implementation of CSR best practices [49], particularly environmental performance [25,129].

Finally, larger companies have better environmental performance [87] because they are more environmentally aware [83]. Therefore, the variable *Temp* was included. This measure is often used in the business performance literature as indicative of firm size [72,116]. Similarly, return on assets (*ROA*) is a critical factor in environmental performance because it influences whether companies will have the resources and capabilities needed for more effective environmental initiatives [130]. Accordingly, several authors, such as Kyaw et al. [25] and Naciti [44], have reported that companies with higher ROA have better environmental performance.

4. Method

To achieve the research aims, as in previous studies [25,27,87,117], econometric estimation of static linear equations was performed. These equations are defined below:

 $EnvSc_{it} = \beta_1 + \beta_2 Gen_{it} + \beta_3 IndepMem_{it} + \beta_4 AveTen_{it} + \beta_5 EnvTra_{it} + \beta_6 Ndir_{it} + \beta_7 Temp_{it} + \beta_8 ROA_{it} + \theta_t + e_{it}$

 $ResUseSc_{it} = \beta_1 + \beta_2 Gen_{it} + \beta_3 IndepMem_{it} + \beta_4 AveTen_{it} + \beta_5 EnvTra_{it} + \beta_6 Ndir_{it} + \beta_7 Temp_{it} + \beta_8 ROA_{it} + \theta_t + e_{it}$ $EmiSc_{it} = \beta_1 + \beta_2 Gen_{it} + \beta_3 IndepMem_{it} + \beta_4 AveTen_{it} + \beta_5 EnvTra_{it} + \beta_6 Ndir_{it} + \beta_7 Temp_{it} + \beta_8 ROA_{it} + \theta_t + e_{it}$ $InnoSc_{it} = \beta_1 + \beta_2 Gen_{it} + \beta_3 IndepMem_{it} + \beta_4 AveTen_{it} + \beta_5 EnvTra_{it} + \beta_6 Ndir_{it} + \beta_7 Temp_{it} + \beta_8 ROA_{it} + \theta_t + e_{it}$

EnvSc is the overall environmental score. *ResUseSc* is the resource use score. *EmiSc* is the emissions score. *InnoSc* is the innovation score. *Gen* corresponds to each of the eight selected gender measures. *IndepMem* refers to the percentage of independent board members. *Ant* is the average number of years for which board members remain on the board. *EnvTra* is a dummy variable that takes the value 1 if employees receive green training, and 0 otherwise. *Ndir* is the number of board members. *Temp* is the number of employees at the company. *ROA* is the return on assets. The term ε_{it} is the random error term for firm *i* in period *t*. Finally, θ_t is a cumulative dummy variable representing time fixed effects. This variable was included to control for unobservable factors that may influence the behaviour of the dependent variable over time.

As these equations are static linear equations based on panel data, they can be estimated using two procedures: a fixed effects model or a random effects model. The choice of model is based on verification of the assumption of the absence of correlation between the unobservable individual effect and the explanatory variables. To do so, the Hausman test is used. If the results of this test lead to the rejection of the hypothesis of absence of correlation, the only consistent estimator is the fixed effects estimator. Conversely, if this hypothesis is not rejected, both estimators are consistent, but the random effects estimator is the efficient estimator [131]. The results of the Hausman test, displayed under "*p* value (Hausman: FE/RE)" in Tables 7–10, reveal the existence of correlation between the explanatory variables and the individual unobservable effect. Therefore, the fixed effects model was used.

The next step required verification of the assumption of exogeneity of the independent variables. This condition is essential for the fixed effects estimator to be consistent [131]. The evidence indicates possible endogeneity problems in the model. This possible endogeneity can be explained by a relationship of bidirectional causality between the dependent variable and the explanatory variables [83,115] or by the omission of relevant variables [35]. Therefore, the Hausman test was used to verify that the hypothesis of zero correlation between the explanatory variables and the error term was met in all cases (exogeneity assumption). The data in Tables 7–10 under "p value (Hausman: FEIV/FE)" reflect this result. For this test, the first lags of the explanatory variables were used as instruments [132].

Finally, estimation was performed using a variance–covariance matrix of errors that were robust to heteroscedasticity between individuals and to serial correlation of the errors of a given individual.

5. Results and Discussion

5.1. Description of the Sample

Before presenting the results of the analysis, this section offers a characterisation of the sample by presenting descriptive statistics for the variables used in the study (Table 6). The measures of interest in this study were the environmental performance and gender diversity indicators.

Variables	Arithmetic Mean	SD	Min	Max	P25	P50	P75	Ν
EnvSc	33.878	30.327	0	96.255	2.122	28.193	62.640	1198
ResUseSc	37.137	33.835	0	99.615	0	33.566	67.171	1198
EmiSc	34.617	34.110	0	99.653	0	26.752	66.463	1198
InnoSc	8.967	22.110	0	97.222	0	0	0	1198
Dum1	0.792	0.406	0	1	1	1	1	1170
Dum2	0.425	0.494	0	1	0	0	1	1170
Dum3	0.289	0.453	0	1	0	0	1	1170
Dum30	0.119	0.325	0	1	0	0	0	1170
Dum40	0.036	0.188	0	1	0	0	0	1170
Nwom	1.504	1.268	0	7	1	1	2	1170
Pwom	0.153	0.118	0	0.571	0.083	0.143	0.222	1170
Blau	0.231	0.151	0	0.5	0.153	0.245	0.345	1170
IndepMem	59.298	22.367	0	100	44.444	62.202	76.923	1174
AveTen	7.61	4.049	0	24.75	4.662	6.853	9.952	1160
EnvTra	0.402	0.49	0	1	0	0	1	1199
Ndir	9.672	2.96	2	26	8	9	11	1194
Temp	37479.5	76565.7	15	508714	3400	11219.5	31000	1318
ROA	7.7607	12	-64.31	204.98	2.719	6.124	10.905	1830

Table 6. Descriptive statistics of the variables (2002–2019).

Source: authors, based on Refinitiv [112]. Compiled using Stata 16, StataCorp LLC, Badajoz, Spain. SD = standard deviation; Min = minimum value; Max = maximum value; P25 = 25th percentile; P50 = 50th percentile; P75 = 75th percentile; N = number of observations.

The statistics relating to the environmental variables reveal low scores overall and by category. The arithmetic mean is below the midpoint score (50) for each of the four measures. For the three categories within the overall index, the highest mean (37.13) corresponds to the reduction of resource use (*ResUseSc*). This value is higher than the corresponding value for the overall score (*EnvSc*). The mean score for the reduction of emissions category (*EmiSc*) is only a few points lower (34.61). The lowest mean value (8.96) is for innovation (*InnoSc*), which is considerably lower than the rest. The four environmental scores have a strikingly wide range, as reflected by the minimum and maximum values. This wide range is consistent with the high dispersion of the observations with respect to the aforementioned means. This dispersion is greater than 30 points for *EnvSc*, *ResUseSc* and *EmiSc*, and is equal to 22.11 points for *InnoSc*.

In line with these data, the percentiles show the poor environmental commitment of the sampled companies and the significant difference between the innovation category and the categories of reduction of resource use and emissions. For example, although 50% of the observations (P50) in the *ResUseSc* and *EmiSc* categories are below 33.56 and 26.75, respectively, 75% of the observations for *InnoSc* are equal to zero (P75).

Analysis of the gender variables shows that some companies did not have women directors in the period analysed (see Min for *Nwom*). Female representation is low and reflects little gender diversity in the sampled firms. Specifically, the average percentage of female directors is only 15.3% (see *Pwom*), with 75% of the observations for this variable taking a value of less than 22.2% (see P75). Although there is at least one case in which women have a representation of 57.1% (see Max for *Pwom*) and therefore outnumber the male directors, the data show that gender parity at the top management level is still a distant goal. For example, female representation on the board is at least 30% for only 11.9% of observations (*Dum30*), and only 3.6% of observations have at least 40% female

representation (*Dum40*). Similar results, which imply a clear under-representation of women on the boards of tourism companies, have already been reported by Barrientos-Báez et al. [133] and Uyar et al. [24]. One possible reason for this finding is that women's career development in the tourism sector is limited, with women finding it difficult to access management positions [16].

With respect to the remaining measures, starting with the governance variables, nearly 60% of board members are independent (*IndepMem*). On average the directors have been on the board for almost 8 years (*AveTen*). Finally, 40.2% of the observations indicate that the sampled tourism companies have environmental training policies (*EnvTra*). The number of employees at these companies ranges massively, from 15 to 508,714 (*Temp*). The sampled tourism companies are profitable, with an average return on assets (*ROA*) of 7.76%. In addition, more than 75% of the observations indicate positive profitability (P25).

In conclusion, despite policies and practices for the reduction of resource use and emissions and for environmental innovation through the development of processes and products designed to help care for the planet, considerable efforts can still be made in all three categories, especially innovation. This situation may stem from a lack of knowledge and interest in environmental protection among directors, as suggested by Erdogan and Baris [134] in their study of a sample of hotels, given that most of the tourism companies analysed in this study have not adopted environmental management training policies (see *EnvTra*, Table 6).

Regarding gender diversity, the under-representation of women on the boards of global tourism companies is notable. The evidence suggests that the tourism sector is some way from attaining the minimum threshold of 40% representation of women on the board recommended by the European Commission in its directive to improve the gender balance among the non-executive directors of listed companies [119].

5.2. Econometric Analysis: Board Gender Diversity and Environmental Performance

This section presents the results of the econometric estimation of the equations stated earlier (Tables 7–10). The aim is to provide statistical evidence of the influence of gender diversity on environmental performance, both generally and specifically in terms of less resource use, lower emissions and environmental innovation. This analysis thus addresses the research aims of the present study. An equation was estimated for each dependent variable and each of the eight proposed gender measures. The models were statistically significant at the 99% confidence level in all cases, as reflected by the *p* value of the F-test, "*p* value (F)".

The relationship of interest in this analysis was the link between gender diversity and environmental performance. Focusing on this relationship, the results indicate a number of interesting findings. First, for the estimates where the dependent variables were *EnvSc*, *ResUseSc* and *EmiSc* (Tables 7–9), four of the eight gender measures (*Dum3*, *Nwom*, *Pwom*, *Blau*) had statistically significant coefficients at a confidence level of 95 or 99%. In all cases, these coefficients were negative. Only in the estimation for *ResUseSc* was the coefficient for *Dum30* statistically significant at the 90% confidence level. This coefficient was also negative. Second, in the case of the dependent variable *InnoSc*, the results clearly differ from those for the other dependent variables. Only the coefficients associated with the variables *Dum30* and *Nwom* were statistically significant at the 90% and 95% confidence level, respectively (Table 10). Both of these coefficients were positive. Finally, none of the coefficients associated with the variables *Dum1*, *Dum2* or *Dum40* were statistically significant (Tables 7–10).

Variables	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Dum1	-1.7955							
Dum2		-2.7040						
Dum3			-4.9474 ***					
Dum30				-3.0859				
Dum40					-4.2413			
Nwom						-2.9679 ***		
Pwom							-31.7088 ***	
Blau								-20.2395 **
IndepMem	-0.0328	-0.0255	-0.0204	-0.0199	-0.0244	-0.0294	-0.0225	-0.0277
AveTen	2.2317 ***	2.2090 ***	2.1423 ***	2.2755 ***	2.2645 ***	2.0953 ***	2.1098 ***	2.1408 ***
EnvTra	15.2397 ***	15.1454 ***	15.097 ***	14.8587 ***	14.9088 ***	14.9823 ***	14.4767 ***	14.6851 ***
Ndir	1.4355 **	1.4437 **	1.3718 **	1.5732 **	1.5383 **	1.8494 ***	1.5794 **	1.5765 **
Temp	-0.9220	-0.7906	-0.7563	-1.2384	-1.1067	-1.0681	-1.1805	-1.0981
ROA	-0.0759	-0.0764	-0.0713	-0.0770	-0.0808	-0.0052	-0.0543	-0.0573
Ν	851	851	851	851	851	851	851	851
R ² (within)	0.6024	0.6035	0.6081	0.6035	0.6030	0.6114	0.6116	0.6090
p value (F)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
<i>p</i> value								
(Hausman:	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
FE/RE)								
<i>p</i> value								
, (Hausman:	0.9977	0.9966	0.9987	0.9962	0.9997	0.8784	0.7285	0.8484
FEIV/FE)								

 Table 7. Dependent variable: EnvSc. Fixed effects estimator (2002–2019).

Source: authors, based on Refinitiv [112]. Compiled using Stata 16, StataCorp LLC, Badajoz, Spain. *** significant at the 99% level, ** significant at the 95% level. R2 (within): coefficient of determination of the transformed model. *p* value (F): *p* value of the test of model significance. *p* value (Hausman: FE/RE): *p* value of the Hausman test under the null hypothesis of absence of correlation between the explanatory variables and the individual unobservable effect. *p* value (Hausman: FEIV/FE): *p* value of the Hausman test under the null hypothesis of absence of correlation between the explanatory variables and the error term. Time dummies are omitted for reasons of brevity and practicality. The estimation was performed with errors that are robust to heteroscedasticity and autocorrelation.

Table 8. Dependent variable: ResUseSc. Fixed effects estimator (2002-2019).

Variables	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Dum1	-1.9426							
Dum2		-2.7762						
Dum3			-5.0060 **					
Dum30				-4.4023 *				
Dum40					-3.9992			
Nwom						-3.0442 ***		
Pwom							-33.1578 ***	
Blau								-20.7839 **
IndepMem	-0.0639	-0.0561	-0.0511	-0.0485	-0.0554	-0.0602	-0.0534	-0.0587
AveTen	2.0655 ***	2.0436 ***	1.9768 ***	2.0968 ***	2.0905 ***	1.9271 ***	1.9267 ***	1.9618 ***
EnvTra	15.9634 ***	15.8663 ***	15.8188 ***	15.6184 ***	15.7519 ***	15.3052 ***	15.2869 ***	15.5114 ***
Ndir	1.3019 **	1.3069 **	1.2331 **	1.4259 **	1.3684 **	1.7230 **	1.4130 **	1.4091 **
Temp	-1.5478	-1.4137	-1.3811	-1.9061	-1.7038	-1.6985	-1.7839	-1.6974
ROA	-0.0755	-0.0764	-0.0712	-0.0752	-0.0811	-0.0511	-0.0533	-0.0568
Ν	851	851	851	851	851	851	851	851
R ² (within)	0.5285	0.5295	0.5336	0.5308	0.5289	0.5370	0.5374	0.5347
p value (F)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
<i>p</i> value								
(Hausman:	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
FE/RE)								
<i>p</i> value								
(Hausman:	0.9942	0.9987	0.9995	0.9995	0.9957	0.9047	0.7756	0.7734
FEIV/FE)								

Source: authors, based on Refinitiv [112]. Compiled using Stata 16, StataCorp LLC, Badajoz, Spain. *** significant at the 99% level, ** significant at the 95% level, * significant at the 90% level. R2 (within): coefficient of determination of the transformed model. *p* value (F): *p* value of the test of model significance. *p* value (Hausman: FE/RE): *p* value of the Hausman test under the null hypothesis of absence of correlation between the explanatory variables and the individual unobservable effect. *p* value (Hausman: FEIV/FE): *p* value of the Hausman test under the null hypothesis of absence of correlation between the explanatory variables and the individual unobservable effect. *p* value (Hausman: FEIV/FE): *p* value of the Hausman test under the null hypothesis of absence of correlation between the explanatory variables and the individual unobservable effect. *p* value (Hausman: FEIV/FE): *p* value of the Hausman test under the null hypothesis of absence of correlation between the explanatory variables and the individual unobservable effect. *p* value (Hausman: FEIV/FE): *p* value of the Hausman test under the null hypothesis of absence of correlation between the explanatory variables and the error term. Time dummies are omitted for reasons of brevity and practicality. The estimation was performed with errors that are robust to heteroscedasticity and autocorrelation.

Variables	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Dum1	-2.0614							
Dum2		-2.8489						
Dum3			-5.4705 **					
Dum30				-3.0401				
Dum40					-6.3484			
Nwom						-3.681 ***		
Pwom							-36.5536 ***	
Blau								-23.0494 **
IndepMem	0.0266	0.0347	0.0404	0.0407	0.0368	0.0304	0.0386	0.0327
AveTen	2.7520 ***	2.7304 ***	2.6541 ***	2.8171 ***	2.7963 ***	2.581 ***	2.6246 ***	2.6624 ***
EnvTra	16.0769 ***	15.9772 ***	15.9190 ***	15.5274 ***	15.511 ***	15.2816 ***	15.0571 ***	15.3022 ***
Ndir	1.7321 **	1.7349 **	1.6590 **	1.9143 **	1.8881 **	2.2511 ***	1.9299 **	1.9259 **
Temp	0.2503	0.3873	0.4329	-0.1170	-0.0025	0.0704	-0.0773	0.0180
ROA	-0.0021	-0.1032	-0.0972	-0.1039	-0.1074	-0.0715	-0.0771	-0.0808
Ν	851	851	851	851	851	851	851	851
R ² (within)	0.5273	0.5281	0.5321	0.5280	0.5286	0.5371	0.5358	0.5333
p value (F)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
<i>p</i> value								
(Hausman:	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
FE/RE)								
p value								
(Hausman:	0.9995	0.9995	0.9995	0.9944	0.9997	0.9629	0.9639	0.9912
FEIV/FE)								

Table 9. Dependent variable: *EmiSc.* Fixed effects estimator (2002–2019).

Source: authors, based on Refinitiv [112]. Compiled using Stata 16, StataCorp LLC, Badajoz, Spain. *** significant at the 99% level, ** significant at the 95% level. R2 (within): coefficient of determination of the transformed model. *p* value (F): *p* value of the test of model significance. *p* value (Hausman: FE/RE): *p* value of the Hausman test under the null hypothesis of absence of correlation between the explanatory variables and the individual unobservable effect. *p* value (Hausman: FEIV/FE): *p* value of the Hausman test under the null hypothesis of absence of correlation between the explanatory variables and the error term. Time dummies are omitted for reasons of brevity and practicality. The estimation was performed with errors that are robust to heteroscedasticity and autocorrelation.

Table 10. Dependent variable: InnoSc. Fixed effects estimator (2002-2019).

Variables	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Dum1	0.5448							
Dum2		-0.6150						
Dum3			-0.3320					
Dum30				6.1967 *				
Dum40					9.2036			
Nwom						2.5081 **		
Pwom							12.8849	
Blau								3.4029
IndepMem	-0.1794 **	-0.1799 *	-0.1801 *	-0.1905 **	-0.1816 **	-0.1799 **	-0.1804 **	-0.1789 *
AveTen	0.0820	0.0636	0.0668	0.0930	0.1182	0.2107	0.1467	0.1000
EnvTra	5.3199 *	5.2999 *	5.3114 *	5.7320 **	5.6523 **	5.8606 **	5.6013 **	5.4328 *
Ndir	0.7953	0.8299 *	0.8138 *	0.7862	0.8537	0.4121	0.8649	0.8789 *
Temp	-1.9699	-1.9323	-1.9534	-1.7054	-1.9651	-1.8544	-1.9879	-2.0265
ROA	0.0799 *	0.0831 *	0.0825 *	0.0724 *	0.0801 *	0.0561	0.0705	0.0775 *
Ν	851	851	851	851	851	851	851	851
R ² (within)	0.2400	0.2401	0.2400	0.2491	0.2474	0.2499	0.2428	0.2409
p value (F)	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
<i>p</i> value								
(Hausman:	0.4520	0.5692	0.4961	0.7653	0.6924	0.9599	0.8575	0.6886
FE/RE)								
<i>p</i> value								
(Hausman:	0.9999	0.9999	0.9999	0.9905	0.9999	0.9997	0.9999	0.9999
FEIV/FE)								

Source: authors, based on Refinitiv [112]. Compiled using Stata 16, StataCorp LLC, Badajoz, Spain. ** significant at the 95% level, * significant at the 90% level. R2 (within): coefficient of determination of the transformed model. *p* value (F): *p* value of the test of model significance. *p* value (Hausman: FE/RE): *p* value of the Hausman test under the null hypothesis of absence of correlation between the explanatory variables and the individual unobservable effect. *p* value (Hausman: FEIV/FE): *p* value of the Hausman test under the null hypothesis of absence of correlation between the explanatory variables and the error term. Time dummies are omitted for reasons of brevity and practicality. The estimation was performed with errors that are robust to heteroscedasticity and autocorrelation.

Therefore, these results provide statistical evidence of a negative influence of board gender diversity on corporate environmental performance. This result is robust because it emerged in several estimates with different dependent variables. Specifically, an increase in the number of women on the board to at least three (*Dum3*), an increase in the number of women on the board (*Nwom*), an increase in the proportion of women (*Pwom*) and greater gender equality among directors (*Blau*) result in a lower environmental performance score (both overall and in terms of reduced resource use and emissions).

According to the estimates for the dependent variable *InnoSc*, the statistical significance of the positive coefficients for *Dum30* and *Nwom* provides evidence that the score for process and product innovation to help care for the planet is greater in companies with at least 30% female representation on the board and increases as the number of women on the board increases.

Third, the statistical evidence from all estimates also indicates no differences in terms of environmental performance between companies without female directors and companies with women on the board (see *Dum1* coefficients), between companies with at least two women on the board and those with fewer than two (see *Dum2* coefficients), and between companies with a minimum representation of 40% women on the board and those that do not meet this threshold (see *Dum40* coefficients).

Consequently, the search for corporate governance mechanisms that influence corporate environmental performance reveals a generally negative relationship in terms of gender diversity on the board of directors. This result is supported by previous studies, which report a decrease in the adoption of environmental practices and environmental policies as the presence of women directors increases [30,31,93,94].

The literature offers several explanations for this negative influence. The first reason is the weak, limited role of women due to their under-representation on the board [135], which is the case in the present sample (Table 6). Second, the presence of women on the board means greater board diversity. This greater diversity may lower firm performance in terms of caring for the planet because of increased conflict and reduced cooperation among board members due to the heterogeneity of the board [40,136]. Finally, the presence of women on the board may be the result of a token gesture solely to improve the corporate image. This situation would limit the scope for women directors to act, preventing their actions from enhancing the firm's environmental performance [30].

However, many studies have shown that companies with greater board gender diversity adopt more actions and strategies aimed at environmental sustainability [24,43,50,73,137]. The argument is that women board members focus more on social and environmental issues [47,138,139]. Therefore, the evidence provided by this study contradicts studies that indicate that board gender diversity is positively related to environmental performance [44,86–88] and reduced resource use and emissions [27,85,140]. These discrepancies can be explained by the complex relationship between board gender diversity and responsible practices [141], which may be determined by the unique characteristics of the context [142]. These characteristics include the company's human resources policy [143], the existence of a CSR committee [144], CEO gender [32], ownership [91] and sector [79,145].

Environmental innovation is the only performance measure with a positive relationship with gender diversity (albeit for only two of the eight estimates). Several arguments support these results. First, there is a stronger relationship between the presence of women and environmental innovation than with the other environmental measures considered in this study. Accordingly, the low female representation does not interfere with the positive effect of women's presence on environmental innovation initiatives [95]. Moreover, the presence of women on the board has certain benefits, such as greater innovation and creativity [60,146]. If these benefits outweigh the internal conflicts and other such negative effects arising from this greater board diversity [147], then environmental innovation may actually be enhanced. This result is supported by several studies, which have shown that companies with greater female board representation are more likely to engage in environmental innovation practices [90] and to implement a greater number of such practices [72,92], especially when the female directors are also independent [91].

This study also supports critical mass theory, with the results confirming the need for there to be at least three women on the board for these female directors to influence environmental actions. In a study of environmental litigation, Liu [115] reported similar findings, although the influence in the present study is negative. This finding can be explained by the fact that the presence of at least one or two women on the board is not enough to ensure that they can influence environmental performance [145]. However, there are discrepancies regarding this theory. For instance, some studies suggest that two women constitute a critical mass. According to these studies, the presence of at least two women on the board is sufficient in order for them to exert a significant influence in several areas, such as the disclosure of information on GHG emissions [58] and environmental innovation [72]. However, Kyaw et al. [25] argued that a critical mass of women on the board is not necessary for better environmental performance, suggesting instead that the presence of one woman on the board is the only relevant factor.

Furthermore, the empirical evidence fails to confirm the need for at least 40% female representation on the board of directors. This percentage has been proposed by the European Parliament and Council Directive [119]. Nevertheless, Fernández-Torres et al. [117] showed that attaining this minimum proportion of women directors is a differentiating factor in corporate performance in the financial sector. However, the results may be due to the low proportion of cases (only 3.6%) where this condition is met (see Table 6). Thus, the desired effect could not be observed.

According to the results, environmental performance is also conditioned by variables other than gender. Specifically, overall environmental performance, resource use and emissions reductions, and environmental innovation are positively influenced by policies to provide environmental management training to employees. Companies with larger boards have better overall environmental performance, as well as adopting a greater number of policies on resource use and emissions reductions. Furthermore, these three environmental performance variables are also positively related to the average tenure of board members. However, tourism companies perform worse in environmental innovation as the percentage of independent board members increases. Finally, higher ROA leads to greater innovation.

To conclude, for the most part, hypotheses *H*1, *H*2 and *H*3 are not supported. The results of the analysis reveal a different relationship from the predicted relationship in the estimates for three of the four dependent variables. However, the significant positive coefficients in the estimates using environmental innovation as the dependent variable provide support for hypothesis *H*4.

6. Conclusions

This study examined 120 listed companies from the global tourism sector over the period 2002 to 2019. The aim was to analyse whether board gender diversity conditions these companies' environmental performance. The analysis, which was based on a fixed effects model, leads to the following conclusions.

The empirical evidence essentially shows that a greater female presence on the board is associated with worse environmental performance. This finding holds for the case of both overall environmental performance and the specific categories of policies and practices related to reducing the use of resources and lowering emissions. The analysis provides robust evidence of this negative influence, which may result from the internal conflicts that arise due to the greater range of opinions and perspectives in gender-diverse groups. Consequently, it is more difficult to reach a consensus on critical decisions. This lack of consensus leads to less effective decision making. Moreover, judging by the low representation of women, any discrepancies that arose in the companies studied would probably be resolved in favour of men (i.e., the dominant group), thus under-utilising female talent. However, the opposite relationship was observed for environmental innovation in two of the eight gender measures. According to this relationship, the performance of tourism firms improves as the number of female board members increases. This finding can be explained by the positive differential effect of female board members on innovation, which may overlap with the above-mentioned conflicts in the decision-making process. This study also supports critical mass theory by showing the need to achieve female representation of at least three women in order for female directors to exert a significant influence on corporate environmental performance.

This research is valuable given the lack of studies that address gender issues in management and their effect on environmental performance in the tourism sector. The findings of this study have practical implications for managers of tourism companies, political decision makers and legislators. First, the under-representation of women on the boards of companies in the tourism sector indicates the existence of a glass ceiling for women in the corporate hierarchy. This limitation has consequences for companies in terms of their environmental performance. This situation highlights the need for regulations in tourism and the ongoing support of diversity education in society to empower women and secure equal conditions for women in the workplace. Specifically, the recommendation is for the implementation of gender quotas at the board level, sanctions for pay gaps between men women, and work-life balance policies to help reconcile work commitments with maternity, childcare and family obligations. Similarly, incentives through tax policies or social recognition should be provided to firms that offer promotions to women. At the same time, tourism managers' commitment to gender diversity is essential. Firms should implement human resource policies based on equal opportunities for men and women that cover all processes, from selection and training to promotion and compensation.

In addition, the lack of involvement of these organisations in caring for the planet is a warning sign for entrepreneurs in this sector, who must increase their awareness and implementation of environmental practices given that their activity depends heavily on the natural resources of the local area. Accordingly, the managers of these firms should create programmes to raise awareness and manage their environmental impact. Such programmes represent the only way to create policies and actions that contribute to corporate environmental education and that build collective awareness and involvement in terms of the need to care for the environment. This care for the environment offers the only way to guarantee access to the resources that this sector depends on for generations to come.

Therefore, there is a need for greater involvement from public administrations, which have a duty to encourage the implementation of these practices through environmental policies and incentives. Such policies should entail sanctions for the abuse of natural resources, the control and reduction of pollution, and the enforcement of reporting on environmental impact and the measures taken to curb this impact. Likewise, improvements in firms' environmental behaviour can be promoted through tax incentives for meeting certain environmental goals or through recognition for best practices in this area, thereby enhancing the corporate image of these firms.

Finally, this study has some limitations. The first limitation relates to the lack of data availability for most of the tourism firms identified in the initial search. This lack of data availability meant that many of these firms were omitted from the study. Another difficulty stems from the use of composite indicators for the environmental performance scores. The use of such indicators can mask important information. Furthermore, the analysis did not control for country effects. Doing so may be important given the cultural diversity and the range of gender and environmental regulations across different countries. Finally, this study did not consider qualitative aspects such as the skills and values of female directors because of the complexity involved in measuring these factors. Analysis of these aspects may help shed light on the relationship between board gender diversity and environmental performance.

One proposal is for future research to use other environmental performance variables that measure more specific elements. Doing so could improve the specification of certain practices. For example, it would be of interest to focus on specific topics addressed in the SDGs relating to environmental issues. The aim would be to obtain indicators that exclusively reflect information on certain items relating to one or more of the targets within a given SDG. Doing so could reveal the involvement of companies in achieving these crucial global goals, as well as providing evidence of the influence of gender diversity in achieving these SDGs. Similarly, it would be of interest to perform analysis with a geographical focus to control for the characteristics of each individual country or region. Finally, other qualitative variables such as the age, education and experience of women managers should be studied as gender variables.

Author Contributions: Conceptualization, Y.F.-T., M.G.-F. and C.G.-S.; Methodology, Y.F.-T., M.G.-F. and C.G.-S.; Software Y.F.-T., M.G.-F. and C.G.-S.; Validation, Y.F.-T., M.G.-F. and C.G.-S.; Formal analysis, Y.F.-T., M.G.-F. and C.G.-S.; Investigation, Y.F.-T., M.G.-F. and C.G.-S.; Resources, Y.F.-T., M.G.-F. and C.G.-S.; Data curation Y.F.-T., M.G.-F. and C.G.-S.; Writing—Original draft preparation, Y.F.-T., M.G.-F. and C.G.-S.; Writing—Review and editing, Y.F.-T., M.G.-F., C.G.-S. and M.G.-F.; funding acquisition, Y.F.-T., M.G.-F. and C.G.-S. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the Junta of Extremadura and was co-financed by the European Regional Development Fund, grant number GR18124. One of the authors is also a beneficiary of a junior university faculty grant (Formación del Profesorado Universitario—FPU) from the Spanish Ministry of Universities, grant number FPU2019-02375.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The data presented in this study are available in the Thomson Reuters Eikon database [112].

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Naciones Unidas. Transformar Nuestro Mundo: La Agenda 2030 Para el Desarrollo Sostenible. Available online: https://www.agenda2030.gob.es/recursos/docs/APROBACION_AGENDA_2030.pdf (accessed on 2 April 2021).
- 2. Soria, J.A.C.; Teigeiro, L.R. The Employment Multiplier in the European Hospitality Industry: A Gender Approach. *Int. J. Contemp. Hosp. Manag.* **2019**, *31*, 105–122. [CrossRef]
- 3. Lasisi, T.T.; Alola, A.A.; Eluwole, K.K.; Ozturen, A.; Alola, U.V. The Environmental Sustainability Effects of Income, Labour Force, and Tourism Development in OECD Countries. *Environ. Sci. Pollut. Res.* 2020, 27, 21231–21242. [CrossRef] [PubMed]
- Jalil, A.; Mahmood, T.; Idrees, M. Tourism-Growth Nexus in Pakistan: Evidence from ARDL Bounds Tests. *Econ. Model.* 2013, 35, 185–191. [CrossRef]
- Habibi, F. The Determinants of Inbound Tourism to Malaysia: A Panel Data Analysis. Curr. Issues Tour. 2017, 20, 909–930. [CrossRef]
- 6. Gössling, S.; Peeters, P.; Ceron, J.-P.; Dubois, G.; Patterson, T.; Richardson, R.B. The Eco-Efficiency of Tourism. *Ecol. Econ.* 2005, 54, 417–434. [CrossRef]
- Zaman, K.; Moemen, M.A.-E.; Islam, T. Dynamic Linkages between Tourism Transportation Expenditures, Carbon Dioxide Emission, Energy Consumption and Growth Factors: Evidence from the Transition Economies. *Curr. Issues Tour.* 2017, 20, 1720–1735. [CrossRef]
- Koçak, E.; Ulucak, R.; Ulucak, Z.Ş. The Impact of Tourism Developments on CO₂ Emissions: An Advanced Panel Data Estimation. *Tour. Manag. Perspect.* 2020, 33. [CrossRef]
- 9. Gössling, S.; Peeters, P. Assessing Tourism's Global Environmental Impact 1900–2050. J. Sustain. Tour. 2015, 23, 639–659. [CrossRef]
- 10. Borja, A.; White, M.P.; Berdalet, E.; Bock, N.; Eatock, C.; Kristensen, P.; Leonard, A.; Lloret, J.; Pahl, S.; Parga, M.; et al. Moving Toward an Agenda on Ocean Health and Human Health in Europe. *Front. Mar. Sci.* **2020**, *7*, 37. [CrossRef]
- 11. Gatti, R.C. Why We Will Continue to Lose Our Battle with Cancers If We Do Not Stop Their Triggers from Environmental Pollution. *Int. J. Environ. Res. Public Health* **2021**, *18*, 6107. [CrossRef] [PubMed]

- Karoui, A.; Crochemore, C.; Harouki, N.; Corbière, C.; Preterre, D.; Vendeville, C.; Richard, V.; Fardel, O.; Lecureur, V.; Vaugeois, J.M.; et al. Nitrogen Dioxide Inhalation Exposures Induce Cardiac Mitochondrial Reactive Oxygen Species Production, Impair Mitochondrial Function and Promote Coronary Endothelial Dysfunction. *Int. J. Environ. Res. Public Health* 2020, *17*, 5526. [CrossRef]
- 13. Saadi, D.; Tirosh, E.; Schnell, I. The Relationship between City Size and Carbon Monoxide (CO) Concentration and Their Effect on Heart Rate Variability (HRV). *Int. J. Environ. Res. Public Health* **2021**, *18*, 788. [CrossRef]
- 14. World Tourism Organization and United Nations Development Programme. *Tourism and the Sustainable Development Goals—Journey to 2030, Highlights;* World Tourism Organization (UNWTO): Madrid, Spain, 2017. [CrossRef]
- 15. Maume, D.J. Is the Glass Ceiling a Unique Form of Inequality? Work Occup. 2004, 31, 250–274. [CrossRef]
- Marinakou, E. Women in Hotel Management and Leadership: Diamond or Glass? *J. Tour. Hosp. Manag.* 2014, *2*, 18–25. [CrossRef]
 Ferguson, L.; Alarcón, D.M. Gender and Sustainable Tourism: Reflections on Theory and Practice. *J. Sustain. Tour.* 2015, *23*, 401–416. [CrossRef]
- 18. Shoham, A.; Almor, T.; Lee, S.M.; Ahammad, M.F. Encouraging Environmental Sustainability through Gender: A Micro-Foundational Approach Using Linguistic Gender Marking. J. Organ. Behav. 2017, 38, 1356–1379. [CrossRef]
- 19. Horbach, J.; Jacob, J. The Relevance of Personal Characteristics and Gender Diversity for (Eco-)Innovation Activities at the Firm-Level: Results from a Linked Employer–Employee Database in Germany. *Bus. Strategy Environ.* **2018**, 27, 924–934. [CrossRef]
- 20. Figueroa-Domecq, C.; Pritchard, A.; Segovia-Pérez, M.; Morgan, N.; Villacé-Molinero, T. Tourism Gender Research: A Critical Accounting. *Ann. Tour. Res.* 2015, *52*, 87–103. [CrossRef]
- 21. Kimbu, A.N.; Ngoasong, M.Z. Women as Vectors of Social Entrepreneurship. Ann. Tour. Res. 2016, 60, 63–79. [CrossRef]
- 22. Cole, S.K.G.; Mullor, E.C.; Ma, Y.; Sandang, Y. "Tourism, Water, and Gender"—An International Review of an Unexplored Nexus. WIREs Water 2020, 7, e1442. [CrossRef]
- 23. Vázquez, R.M.; Barquín, R.D.C.S.; González, M.O.; Cisneros, H.F. Turismo y Género: Una Aproximación Al Estado de Conocimiento. *Gran Tour* 2019, 20, 38–55.
- Uyar, A.; Kilic, M.; Koseoglu, M.A.; Kuzey, C.; Karaman, A.S. The Link among Board Characteristics, Corporate Social Responsibility Performance, and Financial Performance: Evidence from the Hospitality and Tourism Industry. *Tour. Manag. Perspect.* 2020, 35, 100714. [CrossRef]
- 25. Kyaw, K.; Olugbode, M.; Petracci, B. Can Board Gender Diversity Promote Corporate Social Performance? *Corp. Gov.* 2017, 17, 789–802. [CrossRef]
- Lu, J.; Herremans, I.M. Board Gender Diversity and Environmental Performance: An Industries Perspective. *Bus. Strategy Environ*. 2019, 28, 1449–1464. [CrossRef]
- 27. Orazalin, N.; Baydauletov, M. Corporate Social Responsibility Strategy and Corporate Environmental and Social Performance: The Moderating Role of Board Gender Diversity. *Corp. Soc. Responsib. Environ. Manag.* **2020**, *27*, 1664–1676. [CrossRef]
- Martínez-Ferrero, J.; Vaquero-Cacho, L.-A.; Cuadrado-Ballesteros, B.; García-Sánchez, I.M. El Gobierno Corporativo y La Responsabilidad Social Corporativa En El Sector Bancario: El Papel Del Consejo de Administración. *Investig. Eur. Dir. Econ. Empresa* 2015, 21, 129–138. [CrossRef]
- 29. Fakoya, M.B.; Nakeng, M.V. Board Characteristics and Sustainable Energy Performance of Selected Companies in South Africa. *Sustain. Prod. Consum.* **2019**, *18*, 190–199. [CrossRef]
- 30. Deschênes, S.; Rojas, M.; Boubacar, H.; Prud'homme, B.; Ouedraogo, A. The Impact of Board Traits on the Social Performance of Canadian Firms. *Corp. Gov.* 2015, *15*, 293–305. [CrossRef]
- 31. Ardito, L.; Dangelico, R.M.; Petruzzelli, A.M. The Link between Female Representation in the Boards of Directors and Corporate Social Responsibility: Evidence from B Corps. *Corp. Soc. Responsib. Environ. Manag.* **2021**, *28*, 704–720. [CrossRef]
- Birindelli, G.; Iannuzzi, A.P.; Savioli, M. The Impact of Women Leaders on Environmental Performance: Evidence on Gender Diversity in Banks. Corp. Soc. Responsib. Environ. Manag. 2019, 26, 1485–1499. [CrossRef]
- 33. Bear, S.; Rahman, N.; Post, C. The Impact of Board Diversity and Gender Composition on Corporate Social Responsibility and Firm Reputation. *J. Bus. Ethics* **2010**, *97*, 207–221. [CrossRef]
- 34. Galbreath, J. Are There Gender-Related Influences on Corporate Sustainability? A Study of Women on Boards of Directors. J. Manag. Organ. 2011, 17, 017–038. [CrossRef]
- 35. Boulouta, I. Hidden Connections: The Link Between Board Gender Diversity and Corporate Social Performance. *J. Bus. Ethics* **2013**, *113*, 185–197. [CrossRef]
- 36. Pfeffer, J.; Salancik, G.R. *The External Control of Organizations: A Resource Dependence Perspective*; Harper & Row: New York, NY, USA, 1978.
- 37. Pfeffer, J. Merger as a Response to Organizational. *Adm. Sci. Q.* **1972**, *17*, 382–394. [CrossRef]
- Hillman, A.J.; Cannella, A.A.; Paetzold, R.L. The Resource Dependence Role of Corporate Directors: Strategic Adaptation of Board Composition in Response to Environmental Change. J. Manag. Stud. 2000, 37, 235–256. [CrossRef]
- 39. Hillman, A.J.; Cannella, A.A., Jr.; Harris, I.C. Women and Racial Minorities in the Boardroom: How Do Directors Differ? *J. Manag.* **2002**, *28*, 747–763. [CrossRef]
- 40. Pelled, L.H.; Eisenhardt, K.M.; Xin, K.R. Exploring the Black Box: An Analysis of Work Group Diversity, Conflict, and Performance. *Adm. Sci. Q.* **1999**, *44*, 1–28. [CrossRef]
- 41. Daily, C.M.; Dalton, D.R. Women in the Boardroom: A Business Imperative. J. Bus. Strategy 2003, 24, 205–209. [CrossRef]

- 42. Cordeiro, J.J.; Profumo, G.; Tutore, I. Board Gender Diversity and Corporate Environmental Performance: The Moderating Role of Family and Dual-Class Majority Ownership Structures. *Bus. Strategy Environ.* **2020**, *29*, 1127–1144. [CrossRef]
- 43. Feng, X.; Groh, A.; Wang, Y. Board Diversity and CSR. Glob. Financ. J. 2020, 100578. [CrossRef]
- 44. Naciti, V. Corporate Governance and Board of Directors: The Effect of a Board Composition on Firm Sustainability Performance. *J. Clean. Prod.* **2019**, 237, 117727. [CrossRef]
- 45. Freeman, R.E. Strategic Management: A Stakeholder Approach; Pitman Publishing: Boston, MA, USA, 1984.
- 46. Marz, J.W.; Powers, T.L.; Queisser, T. Corporate and Individual Influences on Managers' Social Orientation. J. Bus. Ethics 2003, 46, 1–11. [CrossRef]
- Burkhardt, K.; Nguyen, P.; Poincelot, E. Agents of Change: Women in Top Management and Corporate Environmental Performance. Corp. Soc. Responsib. Environ. Manag. 2020, 27, 1591–1604. [CrossRef]
- 48. Hussain, N.; Rigoni, U.; Orij, R.P. Corporate Governance and Sustainability Performance: Analysis of Triple Bottom Line Performance. J. Bus. Ethics 2018, 149, 411–432. [CrossRef]
- 49. Gulzar, M.A.; Cherian, J.; Hwang, J.; Jiang, Y.; Sial, M.S. The Impact of Board Gender Diversity and Foreign Institutional Investors on the Corporate Social Responsibility (CSR) Engagement of Chinese Listed Companies. *Sustainability* **2019**, *11*, 307. [CrossRef]
- 50. Francoeur, C.; Labelle, R.; Balti, S.; EL Bouzaidi, S. To What Extent Do Gender Diverse Boards Enhance Corporate Social Performance? *J. Bus. Ethics* **2019**, *155*, 343–357. [CrossRef]
- 51. Shahbaz, M.; Karaman, A.S.; Kilic, M.; Uyar, A. Board Attributes, CSR Engagement, and Corporate Performance: What Is the Nexus in the Energy Sector? *Energy Policy* **2020**, *143*, 111582. [CrossRef]
- 52. Eagly, A.H. Sex Differences in Social Behavior; Psychology Press: New York, NY, USA, 1987. [CrossRef]
- 53. Eagly, A.H.; Johnson, B.T. Gender and Leadership Style: A Meta-Analysis. Psychol. Bull. 1990, 108, 233–256. [CrossRef]
- 54. Carli, L.L.; Eagly, A.H. Women Face a Labyrinth: An Examination of Metaphors for Women Leaders. *Gend. Manag.* **2016**, *31*, 514–527. [CrossRef]
- 55. Ravasi, D.; Schultz, M. Responding to Organizational Identity Threats: Exploring the Role of Organizational Culture. *Acad. Manag. J.* **2006**, *49*, 433–458. [CrossRef]
- 56. Rosener, J.B. America's Competitive Secret: Women Managers; Oxford University Press: New York, NY, USA, 1995.
- 57. Frias-Aceituno, J.V.; Rodriguez-Ariza, L.; Garcia-Sanchez, I.-M. The Role of the Board in the Dissemination of Integrated Corporate Social Reporting. *Corp. Soc. Responsib. Environ. Manag.* **2013**, *20*, 219–233. [CrossRef]
- 58. Ben-Amar, W.; Chang, M.; McIlkenny, P. Board Gender Diversity and Corporate Response to Sustainability Initiatives: Evidence from the Carbon Disclosure Project. *J. Bus. Ethics* **2017**, *142*, 369–383. [CrossRef]
- 59. Kanter, R.M. Some Effects of Proportions on Group Life: Skewed Sex Ratios and Responses to Token Women. *Am. J. Sociol.* **1977**, *82*, 965–990. [CrossRef]
- 60. Torchia, M.; Calabrò, A.; Huse, M. Women Directors on Corporate Boards: From Tokenism to Critical Mass. J. Bus. Ethics 2011, 102, 299–317. [CrossRef]
- 61. Yoder, J.D. Rethinking Tokenism: Looking Beyond Numbers. Gend. Soc. 1991, 5, 178–192. [CrossRef]
- 62. Yarram, S.R.; Adapa, S. Board Gender Diversity and Corporate Social Responsibility: Is There a Case for Critical Mass? J. Clean. Prod. 2021, 278, 123319. [CrossRef]
- 63. Fernandez-Feijoo, B.; Romero, S.; Ruiz-Blanco, S. Women on Boards: Do They Affect Sustainability Reporting? *Corp. Soc. Responsib. Environ. Manag.* 2014, 21, 351–364. [CrossRef]
- 64. Cook, A.; Glass, C. Women on Corporate Boards: Do They Advance Corporate Social Responsibility? *Hum. Relations* **2018**, *71*, 897–924. [CrossRef]
- Lin, T.-L.; Liu, H.-Y.; Huang, C.-J.; Chen, Y.-C. Ownership Structure, Board Gender Diversity and Charitable Donation. *Corp. Gov.* 2018, 18, 655–670. [CrossRef]
- 66. Dahlerup, D. From a Small to a Large Minority: Women in Scandinavian Politics. *Scand. Political Stud.* **1988**, 11, 275–298. [CrossRef]
- 67. Joecks, J.; Pull, K.; Vetter, K. Gender Diversity in the Boardroom and Firm Performance: What Exactly Constitutes a "Critical Mass?". J. Bus. Ethics **2013**, 118, 61–72. [CrossRef]
- 68. Konrad, A.M.; Kramer, V.; Erkut, S. Critical Mass: The Impact of Three or More Women on Corporate Boards. *Organ. Dyn.* **2008**, 37, 145–164. [CrossRef]
- 69. Naveed, K.; Voinea, C.L.; Ali, Z.; Rauf, F.; Fratostiteanu, C. Board Gender Diversity and Corporate Social Performance in Different Industry Groups: Evidence from China. *Sustainability* **2021**, *13*, 3142. [CrossRef]
- Post, C.; Rahman, N.; Rubow, E. Green Governance: Boards of Directors' Composition and Environmental Corporate Social Responsibility. *Bus. Soc.* 2011, 50, 189–233. [CrossRef]
- Nuber, C.; Velte, P. Board Gender Diversity and Carbon Emissions: European Evidence on Curvilinear Relationships and Critical Mass. Bus. Strategy Environ. 2021, 30, 1958–1992. [CrossRef]
- 72. He, X.; Jiang, S. Does Gender Diversity Matter for Green Innovation? Bus. Strategy Environ. 2019, 28, 1341–1356. [CrossRef]
- 73. Xie, J.; Nozawa, W.; Managi, S. The Role of Women on Boards in Corporate Environmental Strategy and Financial Performance: A Global Outlook. *Corp. Soc. Responsib. Environ. Manag.* **2020**, *27*, 2044–2059. [CrossRef]
- Wehrmeyer, W.; McNeil, M. Activists, Pragmatists, Technophiles and Tree-Huggers? Gender Differences in Employees' Environmental Attitudes. J. Bus. Ethics 2000, 28, 211–222. [CrossRef]

- 75. Amankwah-Amoah, J.; Danso, A.; Adomako, S. Entrepreneurial Orientation, Environmental Sustainability and New Venture Performance: Does Stakeholder Integration Matter? *Bus. Strategy Environ.* **2019**, *28*, 79–87. [CrossRef]
- Uyar, A.; Kuzey, C.; Kilic, M.; Karaman, A.S. Board Structure, Financial Performance, Corporate Social Responsibility Performance, CSR Committee, and CEO Duality: Disentangling the Connection in Healthcare. *Corp. Soc. Responsib. Environ. Manag.* 2021. [CrossRef]
- 77. Gallego-Sosa, C.; Gutiérrez-Fernández, M.; Fernández-Torres, Y.; Nevado-Gil, M.T. Corporate Social Responsibility in the European Banking Sector: Commitment to the 2030 Agenda and Its Relationship with Gender Diversity. *Sustainability* **2021**, *13*, 1731. [CrossRef]
- 78. Gaard, G. Ecofeminism and Climate Change. Women's Stud. Int. Forum 2015, 49, 20–33. [CrossRef]
- 79. Li, J.; Zhao, F.; Chen, S.; Jiang, W.; Liu, T.; Shi, S. Gender Diversity on Boards and Firms' Environmental Policy. *Bus. Strategy Environ.* **2017**, *26*, 306–315. [CrossRef]
- Nadeem, M.; Zaman, R.; Saleem, I. Boardroom Gender Diversity and Corporate Sustainability Practices: Evidence from Australian Securities Exchange Listed Firms. J. Clean. Prod. 2017, 149, 874–885. [CrossRef]
- 81. Braun, P. Going Green: Women Entrepreneurs and the Environment. Int. J. Gend. Entrep. 2010, 2, 245–259. [CrossRef]
- Post, C.; Rahman, N.; McQuillen, C. From Board Composition to Corporate Environmental Performance Through Sustainability-Themed Alliances. J. Bus. Ethics 2015, 130, 423–435. [CrossRef]
- Kassinis, G.; Panayiotou, A.; Dimou, A.; Katsifaraki, G. Gender and Environmental Sustainability: A Longitudinal Analysis. Corp. Soc. Responsib. Environ. Manag. 2016, 23, 399–412. [CrossRef]
- 84. Glass, C.; Cook, A.; Ingersoll, A.R. Do Women Leaders Promote Sustainability? Analyzing the Effect of Corporate Governance Composition on Environmental Performance. *Bus. Strategy Environ.* **2016**, *25*, 495–511. [CrossRef]
- 85. Biswas, P.K.; Mansi, M.; Pandey, R. Board Composition, Sustainability Committee and Corporate Social and Environmental Performance in Australia. *Pac. Account. Rev.* **2018**, *30*, 517–540. [CrossRef]
- 86. Elmagrhi, M.H.; Ntim, C.G.; Elamer, A.A.; Zhang, Q. A Study of Environmental Policies and Regulations, Governance Structures, and Environmental Performance: The Role of Female Directors. *Bus. Strategy Environ.* 2019, *28*, 206–220. [CrossRef]
- 87. Martín, C.J.G.; Herrero, B. Do Board Characteristics Affect Environmental Performance? A Study of EU Firms. *Corp. Soc. Responsib. Environ. Manag.* 2020, 27, 74–94. [CrossRef]
- 88. Orazalin, N.; Mahmood, M. Toward Sustainable Development: Board Characteristics, Country Governance Quality, and Environmental Performance. *Bus. Strategy Environ.* 2021. [CrossRef]
- Atif, M.; Hossain, M.; Alam, M.S.; Goergen, M. Does Board Gender Diversity Affect Renewable Energy Consumption? J. Corp. Financ. 2020, 66, 101665. [CrossRef]
- 90. Galia, F.; Zenou, E.; Ingham, M. Board Composition and Environmental Innovation: Does Gender Diversity Matter? *Int. J. Entrep. Small Bus.* **2015**, *24*, 117–141. [CrossRef]
- Liao, Z.; Zhang, M.; Wang, X. Do Female Directors Influence Firms' Environmental Innovation? The Moderating Role of Ownership Type. Corp. Soc. Responsib. Environ. Manag. 2019, 26, 257–263. [CrossRef]
- 92. Nadeem, M.; Bahadar, S.; Gull, A.A.; Iqbal, U. Are Women Eco-Friendly? Board Gender Diversity and Environmental Innovation. *Bus. Strategy Environ.* 2020, 29, 3146–3161. [CrossRef]
- 93. Reyes-Bastidas, C.; Briano-Turrent, G.D.C. Las Mujeres en Posiciones de Liderazgo y la Sustentabilidad Empresarial: Evidencia en Empresas Cotizadas de Colombia y Chile. *Estud. Gerenciales* **2018**, *34*, 385–398. [CrossRef]
- 94. Shu, P.-G.; Chiang, S.-J. The Impact of Corporate Governance on Corporate Social Performance: Cases from Listed Firms in Taiwan. *Pac. Basin Financ. J.* 2020, *61*, 101332. [CrossRef]
- Gallego-Sosa, C.; Fernández-Torres, Y.; Gutiérrez-Fernández, M. Does Gender Diversity Affect the Environmental Performance of Banks? Sustainability 2020, 12, 10172. [CrossRef]
- 96. García-Sánchez, I.M.; Gallego-Álvarez, I.; Zafra-Gómez, J.L. Do Independent, Female and Specialist Directors Promote Eco-Innovation and Eco-Design in Agri-Food Firms? *Bus. Strategy Environ.* **2021**, *30*, 1136–1152. [CrossRef]
- 97. Karim, S.; Manab, N.A.; Ismail, R.B. The Dynamic Impact of Board Composition on CSR Practices and Their Mutual Effect on Organizational Returns. *J. Asia Bus. Stud.* **2020**, *14*, 463–479. [CrossRef]
- Govindan, K.; Kilic, M.; Uyar, A.; Karaman, A.S. Drivers and Value-Relevance of CSR Performance in the Logistics Sector: A Cross-Country Firm-Level Investigation. *Int. J. Prod. Econ.* 2021, 231, 107835. [CrossRef]
- 99. Nguyen, T.H.H.; Elmagrhi, M.H.; Ntim, C.G.; Wu, Y. Environmental Performance, Sustainability, Governance and Financial Performance: Evidence from Heavily Polluting Industries in China. *Bus. Strategy Environ.* **2021**, 1–19. [CrossRef]
- 100. Bernardi, R.A.; Threadgill, V.H. Women Directors and Corporate Social Responsibility. *Electron. J. Bus. Ethics Organ. Stud.* **2010**, *15*, 15–21.
- Galbreath, J. Corporate Governance Practices That Address Climate Change: An Exploratory Study. Bus. Strategy Environ. 2010, 19, 335–350. [CrossRef]
- Walls, J.L.; Berrone, P.; Phan, P.H. Corporate Governance and Environmental Performance: Is There Really a Link? *Strategy Manag. J.* 2012, 33, 885–913. [CrossRef]
- Alazzani, A.; Hassanein, A.; Aljanadi, Y. Impact of Gender Diversity on Social and Environmental Performance: Evidence from Malaysia. Corp. Gov. 2017, 17, 266–283. [CrossRef]

- Cuadrado-Ballesteros, B.; García-Sánchez, I.M.; Martínez-Ferrero, J. The Impact of Board Structure on CSR Practices on the International Scale. Eur. J. Int. Manag. 2017, 11, 633–659. [CrossRef]
- Beji, R.; Yousfi, O.; Loukil, N.; Omri, A. Board Diversity and Corporate Social Responsibility: Empirical Evidence from France. J. Bus. Ethics 2020, 1–23. [CrossRef]
- 106. Chen, Z.; Hamilton, T. What Is Driving Corporate Social and Environmental Responsibility in China? An Evaluation of Legacy Effects, Organizational Characteristics, and Transnational Pressures. *Geoforum* 2020, 110, 116–124. [CrossRef]
- 107. Bilimoria, D.; Piderit, S.K. Board Committee Membership: Effects of Sex-Based Bias. Acad. Manag. J. 1994, 37, 1453–1477. [CrossRef]
- 108. Cox, T. The Multicultural Organization. Acad. Manag. Perspect. 2011, 5, 34–47. [CrossRef]
- 109. Goodstein, J.; Gautam, K.; Boeker, W. The Effects of Board Size and Diversity on Strategic Change. *Strategy Manag. J.* **1994**, *15*, 241–250. [CrossRef]
- 110. Diamantopoulos, A.; Schlegelmilch, B.B.; Sinkovics, R.R.; Bohlen, G.M. Can Socio-Demographics Still Play a Role in Profiling Green Consumers? A Review of the Evidence and an Empirical Investigation. *J. Bus. Res.* **2003**, *56*, 465–480. [CrossRef]
- Liao, L.; Luo, L.; Tang, Q. Gender Diversity, Board Independence, Environmental Committee and Greenhouse Gas Disclosure. Br. Account. Rev. 2015, 47, 409–424. [CrossRef]
- 112. Refinitiv. ESG Statement. 2020. Available online: https://eikon.thomsonreuters.com/index.html (accessed on 25 March 2020).
- Refinitiv. Thomson Reuters Business Classification. 2020. Available online: https://www.refinitiv.com/content/dam/marketing/ en_us/documents/quick-reference-guides/trbc-business-classification-quick-guide.pdf (accessed on 30 April 2020).
- 114. Refinitiv. Environmental, Social and Governance (ESG) Scores from Refinitiv. Available online: https://www.refinitiv.com/ content/dam/marketing/en_us/documents/methodology/esg-scores-methodology.pdf (accessed on 30 April 2020).
- 115. Liu, C. Are Women Greener? Corporate Gender Diversity and Environmental Violations. *J. Corp. Financ.* 2018, 52, 118–142. [CrossRef]
- 116. Haque, F. The Effects of Board Characteristics and Sustainable Compensation Policy on Carbon Performance of UK Firms. *Br. Account. Rev.* 2017, 49, 347–364. [CrossRef]
- 117. Torres, Y.F.; Fernández, M.G.; Zurdo, R.J.P. Is Corruption a Determinant of the Effectiveness of Gender Diversity in Business Management? Application to Co-Operative Banks. *Cuad. Gest.* **2020**, *20*, 47–74. [CrossRef]
- 118. Blau, P.M. Inequality and Heterogeneity: A Primitive Theory of Social Structure; Free Press: New York, NY, USA, 1977.
- 119. European Commission. Proposal for a Directive of the European Parliament and of the Council on Improving the Gender Balance among Non-Executive Directors of Companies Listed on Stock Exchanges and Related Measures. 2012. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/PDF/?uri=CELEX:52012PC0614&from=EN (accessed on 30 June 2021).
- 120. Ahern, K.R.; Dittmar, A.K. The Changing of the Boards: The Impact on Firm Valuation of Mandated Female Board Representation. *Q. J. Econ.* **2012**, *127*, 137–197. [CrossRef]
- 121. Lafuente, E.; Vaillant, Y. Balance Rather than Critical Mass or Tokenism: Gender Diversity, Leadership and Performance in Financial Firms. *Int. J. Manpow.* **2019**, *40*, 894–916. [CrossRef]
- 122. Henry, L.A.; Buyl, T.; Jansen, R.J.G. Leading Corporate Sustainability: The Role of Top Management Team Composition for Triple Bottom Line Performance. *Bus. Strategy Environ.* **2019**, *28*, 173–184. [CrossRef]
- 123. Tejerina-Gaite, F.A.; Fernández-Temprano, M.A. The Influence of Board Experience on Firm Performance: Does the Director's Role Matter? J. Manag. Gov. 2020, 25, 685–705. [CrossRef]
- 124. Al-Qahtani, M.; Elgharbawy, A. The Effect of Board Diversity on Disclosure and Management of Greenhouse Gas Information: Evidence from the United Kingdom. *J. Enterp. Inf. Manag.* **2020**, *33*, 1557–1579. [CrossRef]
- 125. Dunn, P.; Sainty, B. The Relationship among Board of Director Characteristics, Corporate Social Performance and Corporate Financial Performance. *Int. J. Manag. Financ.* **2009**, *5*, 407–423. [CrossRef]
- Kassinis, G.; Vafeas, N. Corporate Boards and Outside Stakeholders as Determinants of Environmental Litigation. *Strategy Manag. J.* 2002, 23, 399–415. [CrossRef]
- 127. Barney, J. Firm Resources and Sustained Competitive Advantage. J. Manag. 1991, 17, 99–120. [CrossRef]
- 128. Jensen, M.C. The Modern Industrial Revolution, Exit, and the Failure of Internal Control Systems. J. Financ. 1993, 48, 831–880. [CrossRef]
- 129. De Villiers, C.; Naiker, V.; van Staden, C.J. The Effect of Board Characteristics on Firm Environmental Performance. *J. Manag.* **2011**, *37*, 1636–1663. [CrossRef]
- 130. Zou, H.; Xie, X.; Qi, G.; Yang, M. The Heterogeneous Relationship between Board Social Ties and Corporate Environmental Responsibility in an Emerging Economy. *Bus. Strategy Environ.* **2019**, *28*, 40–52. [CrossRef]
- 131. Pérez López, C. Problemas Resueltos de Econometría; Ediciones Paraninfo S.A.: Madrid, Spain, 2006.
- Arellano, M.; Bond, S. Some Tests of Specification for Panel Data: Monte Carlo Evidence and an Application to Employment Equations. *Rev. Econ. Stud.* 1991, 58, 277–297. [CrossRef]
- 133. Barrientos-Báez, A.B.; Báez-García, A.J.; Flores-Muñoz, F.; Gutiérrez-Barroso, J. Gender Diversity, Corporate Governance and Firm Behavior: The Challenge of Emotional Management. *Eur. Res. Manag. Bus. Econ.* **2018**, *24*, 121–129. [CrossRef]
- 134. Erdogan, N.; Baris, M.E. Environmental Protection Programs and Conservation Practices of Hotels in Ankara, Turkey. *Tour. Manag.* **2007**, *28*, 604–614. [CrossRef]

- 135. Khan, H.U.Z. The Effect of Corporate Governance Elements on Corporate Social Responsibility (CSR) Reporting: Empirical Evidence from Private Commercial Banks of Bangladesh. *Int. J. Law Manag.* **2010**, *52*, 82–109. [CrossRef]
- Williams, K.Y.; O'Reilly, C.A. Demography and Diversity in Organizations: A Review of 40 Years of Research. *Res. Organ. Behav.* 1998, 20, 77–140.
- 137. Wang, Y.; Wilson, C.; Li, Y. Gender Attitudes and the Effect of Board Gender Diversity on Corporate Environmental Responsibility. *Emerg. Mark. Rev.* **2021**, 47, 100744. [CrossRef]
- Zaid, M.A.A.; Wang, M.; Adib, M.; Sahyouni, A.; Abuhijleh, S.T.F. Boardroom Nationality and Gender Diversity: Implications for Corporate Sustainability Performance. J. Clean. Prod. 2020, 251, 119652. [CrossRef]
- 139. Provasi, R.; Harasheh, M. Gender Diversity and Corporate Performance: Emphasis on Sustainability Performance. *Corp. Soc. Responsib. Environ. Manag.* 2021, 28, 127–137. [CrossRef]
- 140. Román, C.C.; Zorio-Grima, A.; Merello, P. Economic Development and CSR Assurance: Important Drivers for Carbon Reporting ... yet Inefficient Drivers for Carbon Management? *Technol. Forecast. Soc. Chang.* **2021**, *163*, 120424. [CrossRef]
- 141. Yang, W.; Yang, J.; Gao, Z. Do Female Board Directors Promote Corporate Social Responsibility? An Empirical Study Based on the Critical Mass Theory. *Emerg. Mark. Financ. Trade* 2019, *55*, 3452–3471. [CrossRef]
- 142. Chang, Y.K.; Oh, W.-Y.; Park, J.H.; Jang, M.G. Exploring the Relationship Between Board Characteristics and CSR: Empirical Evidence from Korea. J. Bus. Ethics 2017, 140, 225–242. [CrossRef]
- Kochan, T.; Bezrukova, K.; Ely, R.; Jackson, S.; Joshi, A.; Jehn, K.; Leonard, J.; Levine, D.; Thomas, D. The Effects of Diversity on Business Performance: Report of the Diversity Research Network. *Hum. Resour. Manag.* 2003, 42, 3–21. [CrossRef]
- 144. Martínez-Ferrero, J.; Eryilmaz, M.; Colakoglu, N. How Does Board Gender Diversity Influence the Likelihood of Becoming a UN Global Compact Signatory? The Mediating Effect of the CSR Committee. *Sustainability* **2020**, *12*, 4329. [CrossRef]
- 145. Wei, F.; Ding, B.; Kong, Y. Female Directors and Corporate Social Responsibility: Evidence from the Environmental Investment of Chinese Listed Companies. *Sustainability* **2017**, *9*, 2292. [CrossRef]
- 146. Griffin, D.; Li, K.; Xu, T. Board Gender Diversity and Corporate Innovation: International Evidence. J. Financ. Quant. Anal. 2021, 56, 123–154. [CrossRef]
- 147. Carter, D.A.; D'Souza, F.; Simkins, B.J.; Simpson, W.G. The Gender and Ethnic Diversity of US Boards and Board Committees and Firm Financial Performance. *Corp. Gov. Int. Rev.* **2010**, *18*, 396–414. [CrossRef]